Author's response to reviews

**Title:** Compulsory and Recommended Vaccination in Italy: evaluation of coverage and non-compliance between 1998-2002 in Northern Italy

**Authors:**

Serena Stampi (serena.stampi@unibo.it)
Rita Ricci (Rita.Ricci@ausl.bo.it)
Isa Ruffilli (Rita.Ricci@ausl.bo.it)
Franca Zanetti (fzanetti@alma.unibo.it)

**Version:** 3 **Date:** 13 April 2005

**Author's response to reviews:**

The manuscript has been revised previously in light of the reviewers comments (see revised version 18 march 2005)

Now: changes made to the format:

1- Title: Northern Italy instead of Bologna
2- List of Authors completed with address etc.
3- Each author has been linked to superscript number
4- Competing interests included
5- Authors’ contribution included
6- Borders of tables : black lines
7- page 18 has been removed